Author’s response to reviews

Title: Paroxysmal extreme pain disorder in family with c.3892G>T (p.Val1298Phe) in the SCN9A gene mutation - case report

Authors:

Adam Stępień (astepien@wim.mil.pl)
Daria Salacińska (dsalacinska@wim.mil.pl)
Jacek Staszewski (jstaszewski@wim.mil.pl)
Marta Durka-Kęsy (mdurka-kesy@wim.mil.pl)
Jan Dobrogowski (jandobrogowski1@gmail.com)

Version: 3 Date: 27 Apr 2020

Author’s response to reviews:

Dear Editor and Reviewers,

Thank you for valuable comments regarding presented manuscript. In revised version we took into account most comments from reviewers.

In response to comments of the Reviewer 2:
1) Unfortunately we do not have access to the details of the genetic test. We only have the result of this tests, so we can not add picture of the chromatogram.
2) Individuals with genetic data is highlighted on the pedigree.
3) Sentences were corrected where it was necessary.
4) We checked and added references where it was necessary.

We also revised the text for language and grammar issues with a native. If there are still language errors, we will send the manuscript to one of professional editing service that you recommend.

We hope that the revised version meets the expectations of reviewers.

Kind regards,
Daria Salacińska